% !TEX TS-program = pdflatexmk
% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P20}{20(2)}{2013}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins
\usepackage{appendix}

%\usepackage{setspace}


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\type}{\mathcal{T}}
\def\RR{{\mathbb R}}
\def\Po{{\mathcal P}}
\def\NN{{\mathbb N}}
\def\ZZ{{\mathbb Z}}
\def\QQ{{\mathbb Q}}
\def\CC{{\mathbb C}}
\def\EE{{\mathbb E}}
\newcommand{\ol}[1]{\overline{#1}}
\newcommand{\oimp}[1]{\overset{#1}{\Longleftrightarrow}}
\newcommand{\bv}[1]{\ensuremath{ \vec{\mathbf{#1}}} }
\newcommand{\mc}[1]{\ensuremath{\mathcal{#1}}}

\usepackage[dvipsnames,usenames]{color}
\newtheorem*{acomment}{\color{BrickRed}{Comment}}
\newcommand{\acom}[2]{{\begin{acomment}[#1]\color{OliveGreen} #2 \color{black} \end{acomment}}}
\newcommand{\red}[1]{{\color{BrickRed} #1}}
\newcommand{\blue}[1]{{\color{MidnightBlue}#1}}
\newcommand{\green}[1]{{\color{OliveGreen} #1}}
\newcommand{\mwrong}[1]{\red{\cancel{#1}}}
\newcommand{\wrong}[2]{\red{\sout{#1}\green{#2}}}
\definecolor{OliveGreen}{rgb}{.3,.5,.2}
\definecolor{MidnightBlue}{rgb}{.3,.4,.6}
\newcommand{\gcom}[1]{\acom{GW}{#1}}
\newcommand{\mon}[1]{\text{Mon}(\mc #1)}
\newcommand{\cl}[2]{\text{Cl}_{#1}(#2)}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\title{Minimal covers of the Archimedean tilings, part II}
\author{Mark Mixer\\ Daniel Pellicer \\Gordon Williams}
\author{Mark Mixer\\
\small Department of Mathematics and Statistics\\[-0.8ex]
\small Williams College \\[-0.8ex]
\small Williamstown, MA, USA\\
\small\tt mark.mixer@williams.edu
\and
Daniel Pellicer\thanks{partially supported by PAPIIT--Mexico under grant IB100312 and CONACYT project 166951}\\
\small Research Center on Mathematical Sciences\\[-0.8ex]
\small National University of Mexico \\[-0.8ex]
\small Morelia, Michoacan, Mexico\\
\small\tt pellicer@matmor.unam.mx
\and Gordon Williams\thanks{partially supported by PAPIIT--Mexico under grant IB100312 \newline
All of the authors would also like to acknowledge the Fields Institute's 2011 Workshop on Symmetry in Graphs, Maps and Polytopes.}\\
\small Department of Mathematics and Statistics\\[-0.8ex]
\small University of Alaska Fairbanks\\[-0.8ex]
\small Fairbanks, Alaska, USA\\
\small\tt gwilliams@alaska.edu
}

%\thanks{The authors would also like acknowledge the opportunity to collaborate we were given by the Fields Institute Workshop on Symmetry in Graphs, Maps and Polytopes, 2011.}

\date{\dateline{Jan 24, 2013}{Apr 12, 2013}{Apr 24, 2013}\\
\small Mathematics Subject Classifications: 52B15, 52B70, 52C20}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}
\begin{thebibliography}{10}
\bibitem[Bra27]{Bra27}
H.~R. Brahana, \emph{Regular maps and their groups}, Amer. J. Math. \textbf{49}
  (1927), no.~2, 268--284.

\bibitem[CM80]{CoxMos80}
H.~S.~M. Coxeter and W.~O.~J. Moser, \emph{Generators and relations for
  discrete groups}, fourth ed., Ergebnisse der Mathematik und ihrer
  Grenzgebiete [Results in Mathematics and Related Areas], vol.~14,
  Springer-Verlag, Berlin, 1980. %\MR{562913 (81a:20001)}

\bibitem[CPR+]{LochNess}
Thierry Coulbois, Daniel Pellicer, Miguel Raggi, Camilo Ram{\'\i}rez, and
  Ferr{\'a}n Valdez, \emph{The topology of the minimal regular cover of the
  archimedean tessellations}, (in review).

\bibitem[DM13]{DraMix13}
Kostiantyn Drach and Mark Mixer, \emph{Minimal covers of equivelar toroidal
  maps}, (in review) (2013).

\bibitem[DS]{DraSkor13}
Kostiantyn Drach and Maksym Skoryk, \emph{Archimedean toroids and their minimal
 almost regular covers.}, (in preparation).

\bibitem[GAP]{GAP}
M~Sch\"onert et~al., \emph{Gap: Groups, algorithms and programming}, Lehrstuhl
  D f\"ur Mathematik, Aachen (1994).

\bibitem[GS87]{GruShe87}
Branko Gr{\"u}nbaum and G.~C. Shephard, \emph{Tilings and patterns}, W. H.
  Freeman and Company, New York, 1987.

\bibitem[Har99a]{Har99}
Michael~I. Hartley, \emph{All polytopes are quotients, and isomorphic polytopes
  are quotients by conjugate subgroups}, Discrete Comput. Geom. \textbf{21}
  (1999), 289--298.

\bibitem[Har99b]{Har99a}
Michael~I. Hartley, \emph{More on quotient polytopes}, Aequationes Math. \textbf{57}
  (1999), 108--120.

\bibitem[HOW09]{HubOrbWei09}
Isabel Hubard, Alen Orbani\'c, and A.~Ivic Weiss, \emph{Monodromy groups and
  self-invariance}, Canad. J. Math. \textbf{61} (2009), 1300--1324.

\bibitem[HW10]{HarWil10} Michael~I. Hartley and Gordon~I. Williams, \emph{Representing the sporadic {A}rchimedean polyhedra as abstract polytopes}, Discrete Math. \textbf{310}  (2010), no.~12, 1835--1844.

\bibitem[MPW]{MonPelWil11a}
Barry Monson, Daniel Pellicer, and Gordon~I. Williams, \emph{Mixing and
  monodromy of abstract polytopes}, Trans. of the AMS \textbf{(in press)}.

\bibitem[MS02]{McMSch02}
Peter McMullen and Egon Schulte, \emph{Abstract regular polytopes}, Cambridge
  University Press, 2002.

\bibitem[Orb07]{Orb07}
Alen Orbani{{\'c}}, \emph{{$F$}-actions and parallel-product decomposition of
  reflexible maps}, J. Algebraic Combin. \textbf{26} (2007), no.~4, 507--527.

\bibitem[PW11a]{PelWei11}
Daniel Pellicer and Asia~Ivi\'c Weiss, \emph{Uniform maps on surfaces of
  non-negative {E}uler characteristic}, Symmetry \textbf{22} (2011), no.~1-2,
  159--196.

\bibitem[PW11b]{PelWil10}
Daniel Pellicer and Gordon~I. Williams, \emph{Quotient representations of
  uniform tilings}, Symmetry Cult. Sci. \textbf{22} (2011), no.~3-4, 289--306.

\bibitem[PW12]{PelWil12}
Daniel Pellicer and Gordon~I. Williams, \emph{Minimal covers of the {A}rchimedean tilings, part 1}, Electron.
  J. Combin. \textbf{19} (2012), no.~3, \#P6.

\bibitem[Sch03]{Sch03}
Egon Schulte, \emph{ Combinatorial aperiodicity of polyhedral prototiles},
in {\em Discrete geometry}, volume 253 of {\em Monogr. Textbooks Pure
  Appl. Math.}, pages 397--406. Dekker, New York, 2003.


\bibitem[Wil76]{Wil76}
Stephen~E. Wilson, \emph{New techniques for the construction of regular maps},
  Ph.D. thesis, University of Washington, 1976.

\bibitem[Wil94]{Wil94}
Stephen~E. Wilson, \emph{Parallel products in groups and maps}, J. Algebra \textbf{167}
  (1994), no.~3, 539--546. %\MR{1287058 (95d:20067)}

\bibitem[Wil02]{Wil02a}
Stephen~E. Wilson, \emph{Families of regular graphs in regular maps}, J. Combin. Theory
  Ser. B \textbf{85} (2002), no.~2, 269--289. %\MR{1912968 (2003d:05058)}


\end{thebibliography}


\maketitle

\begin{abstract}
In Part I of this paper (\cite{PelWil12}), the minimal regular covers of three  of the eight Archimedean tilings were determined.  However, the computations described in that work grow more complicated as the number of flag orbits of the tilings increases.  In Part II, we develop a new technique in order to present the minimal regular covers of certain periodic abstract polytopes.  We then use that technique to finish determining the minimal regular covers of the Archimedean tilings.
\end{abstract}

%\acom{DP}{Gordon, you are supposed to thank the same grant I did. Your support last year came from there}
%\acom{MM}{Gordon, do you want/need the \{ \} around your email?}
%\acom{MM}{I have cleaned up the appendices as best I could.}
%\acom{DP}{I hope that's enough.}
%\gcom{I have had reviewers make snide remarks in the past about having citations appear in the abstract.}
%\acom{DP}{I feel natural to cite the first part in the abstract. If a referee complains, perhaps he will suggest a wording to avoid it. In any case, that shouldn't be a factor for the paper's acceptance}
%\gcom{Also, I now have the file in the EJC style (mostly, I think). But this has introduced some weirdness with the tables at the end. Anyone have any ideas how to fix?}
%\acom{DP}{I don't think we have to. First of all, we should see how it goes once all the comments are gone. Second, we should worry about it when it is about to go to the editorial process; so if some corrections are suggested, we fix them without spoiling any previous effort on having the tables in the correct place. We just need to cite and caption properly those tables.}
\section{Introduction}\label{sec:introd}


Symmetric maps on surfaces, and their automorphism groups, have been studied since the early 20th century, mainly in the case when the maps are finite, that is, when they lie on compact surfaces (see \cite{Bra27}, \cite{CoxMos80}). Much is known about the regular covers and the closely related monodromy groups of such finite maps (e.g., \cite{Orb07,Wil76, Wil94, Wil02a}).  However, tilings of the Euclidean or hyperbolic plane are examples of maps on non-compact surfaces, and as such, require somewhat different methods of investigation.

Since the late 20th century, maps whose vertex-figures are polygons have been studied as abstract polyhedra (rank 3 abstract polytopes) \cite[Section 6B]{McMSch02}.
Much work has been done to understand (abstract) regular polytopes, where \cite{McMSch02} is the primary reference work. However, in comparison, much less is known about polytopes with less symmetry.  A particularly useful tool for the study of non-regular abstract polytopes is \cite{Har99}, in which Hartley demonstrates that any abstract polytope can be realized as the quotient of a regular (abstract) polytope. There is a strong connection between this idea and the concept of regular covers of maps. In particular, every map has a \textit{minimal regular cover}, in the sense that no other regular cover  is covered by the minimal one. It is easy to demonstrate that the minimal regular cover of a polyhedron is itself a polyhedron (details are provided in \cite{MonPelWil11a}). This paper and its predecessor \cite{PelWil12} are part of a project whose goal is to better understand the structure of  quotients and covers both geometrically and algebraically.  For a discussion of topological aspects of the covers of infinite maps see~\cite{LochNess}.


Symmetric tilings have been the subject of intense interest and investigation, especially since the publication of Gr\"unbaum and Shephard's {\em Tilings and Patterns} \cite{GruShe87}.  Consequently, the Archimedean tilings provide a natural starting point in the study of regular covers of infinite maps.  In~\cite{PelWil10} representations of the Archimedean tilings as quotients of regular hyperbolic tilings were provided by means of a general description developed by Hartley in~\cite{Har99} (more basic information about such representations is available in~\cite{Har99a} and~\cite{HarWil10}). The minimal regular covers of the three Archimedean tilings (3.6.3.6), (4.8.8) and (3.12.12) were found in  \cite{PelWil12}. These tilings have in common that the number of orbits on flags under the symmetry group is at most 3. The technique used consists of finding the intersection of stabilizers of certain flags under the action of the monodromy group of the tiling.  While this method is very general, and can be used to work with arbitrary abstract polytopes, the calculations were lengthy and seemed intractable for the remaining five Archimedean tilings.


Using new techniques, in this paper we complete the determination of the minimal regular covers of the Archimedean tilings.  The theory leading to the procedure which we used to find the minimal regular covers of the Archimedean tilings (3.4.6.4), (3.3.4.3.4), (3.3.3.4.4), (4.6.12) and (3.3.3.3.6) is provided in Section \ref{sec:structure}. This procedure is a natural tool which could be applied to other periodic tessellations of Euclidean spaces.  In Section \ref{sec:3464} we show the details of this procedure while considering the tiling (3.4.6.4). Finally, an application of this procedure to finite maps on the torus is discussed in Section~\ref{sec:torus}.


\section{Basic notions and previous results}\label{sec:notions}
%{\bf Gordon and Daniel}

In this paper, we define {\em polyhedra} as abstract
polytopes of rank $3$ in the sense of \cite{McMSch02}; that is, a polyhedron is a partially ordered set with a rank function on the set $\{0,1,2\}$ satisfying certain properties. Following the notation of convex polyhedra, the {\em vertices}, {\em edges} and {\em faces} of an abstract polyhedron are its
elements of rank $0$, $1$ and $2$ respectively. Every vertex belongs to at least two edges, and every face contains at least two edges.

A {\em section} $F/G$ of a polyhedron is defined when $G\le F$ to be the partially ordered set of all elements $H$ such that $G\le H\le F$, that is $F/G:=\left\{H\in \mc P \mid G\le H\le F\right\}$. The {\em vertex-figure}
at a vertex $v$ is the section $\{F \in \mc P\, |\, v \le F\}$. Abstract polyhedra satisfy the {\em diamond condition}, that is, every edge contains precisely two vertices
and is contained in precisely two faces, and the section determined by an incident face $f$ and  vertex $v$ pair contains exactly two edges.
A {\em flag} is a (vertex, edge, face) triple of mutually incident elements of the partially ordered set.
%\acom{MM}{We forgot to define a flag, and the definition of a section used the word face in a misleading way}
As a consequence of the diamond condition, given $i \in \{0, 1, 2\}$ and a flag $\Psi$,
there exists a unique flag $\Psi^i$ that coincides with $\Psi$ in all elements except
in that of rank $i$. The flag $\Psi^i$ is called the {\em i-adjacent flag} of
$\Psi$. Finally, an abstract polyhedron is also {\em strongly flag-connected}, that is, any two flags in a polyhedron can be connected by a sequence of adjacent flags, and this connectivity is preserved when restricted to either faces or vertex-figures.
%\gcom{This originally said ``Finally, abstract polyhedra are also {\em strongly flag-connected}, that is, each face and each
%vertex-figure is isomorphic to an abstract polygon (the partially ordered set induced by a convex polygon).'' I don't think this definition even forces the polyhedron to be connected. I've inserted my own suggestion for how to fix it.}

The {\em degree} of a vertex $v$ is the number of edges containing $v$, and the {\em co-degree} of a face $f$ is the number of edges contained in $f$.
 Whenever every vertex of a polyhedron $\Po$ has the same degree $q$, and every face of
$\Po$ has the same co-degree $p$ we say that $\Po$ is {\em equivelar} and has {\em Schl\"afli
type} $\{p, q\}$.
%\acom{MM}{We had this backwards before.}

An {\em automorphism} of a polyhedron $\Po$ is an order preserving bijection of its elements. We say that a polyhedron is {\em regular} if its automorphism group $\Gamma(\Po)$ acts transitively on the set of flags of \mc P. The Platonic solids and the familiar regular tessellations of the Euclidean plane by triangles, squares, or hexagons ($3^6$, $4^4$, and $6^3$ in the notation of \cite{GruShe87}) are examples of abstract regular polyhedra.

   A {\em string C-group} $G$ of rank $3$ is a group which %is not dihedral and
has distinct distinguished
involutory generators $\rho_0, \rho_1, \rho_2$ satisfying that
$(\rho_0 \rho_2)^2 = id$, the identity in $G$, and that $\langle \rho_0, \rho_1 \rangle \cap \langle \rho_1, \rho_2 \rangle
= \langle \rho_1 \rangle$  (the latter is called the {\em intersection condition}). The automorphism group of an abstract regular polyhedron \mc P is always a string
C-group of rank $3$. Given an arbitrarily chosen {\em base flag} $\Phi$ of \mc P, $\rho_{i}$ can be taken to be the (unique) automorphism mapping $\Phi$ to the $i$-adjacent flag $\Phi^{i}$.
Furthermore, any string C-group $\Gamma$ of rank $3$ is the automorphism
group of an abstract regular
polyhedron $\mathcal{P}(\Gamma)$ ~\cite[2B]{McMSch02}, so, up to isomorphism, there is a one-to-one correspondence between the string C-groups of rank $3$
and the abstract regular polyhedra.  Thus, in the study of regular abstract polyhedra we may either  work with the polyhedron as a poset or with its automorphism group. %In this paper the automorphism group of a regular polyhedron $\Po$ will be denoted by $\Gamma(\Po)$.

Examples of regular polyhedra are the regular tilings. We define a {\em tiling} of the (spherical, Euclidean or hyperbolic) plane as an edge to edge covering of the plane by convex  polygons whose interiors are pairwise disjoint. There are three {\em regular tilings} of the Euclidean plane, namely the tessellations by equilateral triangles, squares and regular hexagons.

   For any polyhedron $\Po$ we define permutations $r_0, r_1, r_2$ on $\mc F(\mc P)$ by
\[\Psi{r_i} := \Psi^i,\]
for every flag $\Psi$ of $\Po$ and $i=0, 1, 2$ (note that these are {\em not} automorphisms of \mc P). The group $\mon{\mc P}:=\langle r_0, r_1, r_2 \rangle$
will be referred to as the {\em monodromy group} of $\Po$ ~\cite{HubOrbWei09}.  It is now well known that $\mon{\Po}$ is a string C-group for every polyhedron $\Po$ (see for example ~\cite{MonPelWil11a}).

Note that, by definition of automorphism, the action of each $r_i$ commutes with the automorphisms of any
given polyhedron. That is,
\begin{equation}\label{eq:actionsymmetry}
\alpha (\Psi{r_i}) = (\alpha \Psi){r_i}
\end{equation}
for $i=0, 1, 2$ and $\alpha \in \Gamma(\Po)$. More generally, an inductive process shows that the action of the monodromy group on the flags commutes with the action by automorphisms, that is,
\begin{equation}\label{eq:actions2}
\alpha (\Psi{w}) = (\alpha \Psi){w}
\end{equation}
for every $w \in \mon{\Po}, \alpha \in \Gamma(\Po)$. In other words, the left action of $\Gamma(\Po)$ and right action of $\mon{\Po}$ on the set of flags are compatible.

The {\em flag action} of a string C-group $\Gamma=\langle \rho_0, \rho_1, \rho_2 \rangle$ on $\Po$ is the group homomorphism
$\Gamma\to \mon{\mc P}$ defined by $\rho_i\mapsto r_i$, provided such a homomorphism exists.
We say that the regular polyhedron $\mc P$ is a {\em cover} of $\mc Q$, denoted by $\mc P\searrow \mc Q$, if
$\mc Q$ admits a flag action from $\Gamma(\mc P)$.
%\acom{MM}{Do we want to define non-regular covers?}
For example, the (universal) polyhedron with automorphism group isomorphic to the
Coxeter group
$[\infty, \infty] := \langle \rho_0, \rho_1, \rho_2 \,|\, (\rho_0 \rho_2)^2 = id \rangle$ covers all other polyhedra. Whenever the least common multiple of the co-degrees
of the faces of a polyhedron $\mc P$ is $p$, and the least common multiple of the vertex degrees  of $\mc P$ is $q$, $\mc P$ is covered by the
tiling $\{p, q\}$ whose automorphism group is isomorphic to the string Coxeter group
\[[p, q] := \langle \rho_0, \rho_1, \rho_2 \,|\, (\rho_0 \rho_2)^2 =
(\rho_0 \rho_1)^p = (\rho_1 \rho_2)^q = id \rangle.\]
Recall that $\{p,q\}$ can be viewed as a regular tiling of the sphere, Euclidean plane or hyperbolic plane, depending on whether $\frac{1}{p}+\frac{1}{q}$ is bigger than, equal to, or less than $\frac{1}{2}$, respectively.

Whenever $\mc P\searrow \mc Q$,  the polyhedron $\mc Q$ is totally
determined by $\mc P$ and the stabilizer $N$ of a chosen base flag $\Phi$ of $\mc Q$ under the flag
action of $\Gamma(\mc P)$. Indeed,  $\mc Q = \mc P / N$, the polytope whose faces are orbits under the action of $N$ on $\mc P$ \cite{Har99}.

Since the monodromy group of every polyhedron is a string C-group, any polyhedron ${\mathcal Q}$ is automatically equipped with a regular cover, $\mathcal{P}(\mon{Q})$. Moreover, it is straightforward to demonstrate that
\begin{equation}\label{eq:core}
\mon{\mc Q}\cong \Gamma/\text{Core}(\Gamma,N),
\end{equation}
where $\Gamma$ is the automorphism group of any regular cover of \mc Q, $N$ is the stabilizer of a flag in \mc Q under the flag action of $\Gamma$, and the {\em core}, $\text{Core}(\Gamma,N)$, is the largest normal subgroup of $\Gamma$ in $N$ (see ~\cite{MonPelWil11a} for details). %Note that
%\[\text{Core}(\Gamma,N) = \bigcap_{g \in \Gamma} g^{-1} N g.\]
In particular, it follows from (\ref{eq:core}) that there is a group homomorphism from $\Gamma$ to $\mon{Q}$ mapping the generator $\rho_i$ to $r_i$, implying that any regular cover of $\mathcal{Q}$ also covers the polyhedron $\mathcal{P}(\mon{Q})$. This motivates the name ``minimal regular cover of $\mathcal{Q}$'' for $\mathcal{P}(\mon{Q})$ since it is actually the smallest possible regular cover.

As a consequence of the previous discussion, the monodromy group of a regular polytope $\mc P$ is isomorphic to $\Gamma({\mc P}) / \text{Core}(\Gamma,N)$, where $N$ is the stabilizer of any flag in \mc P. Note that in this case $\text{Core}(\Gamma,N)$ is trivial, since the stabilizer of all flags is the same as the stabilizer of any individual flag (due to regularity) and every conjugate of the stabilizer $N$ of a flag of $\mc P$ is the stabilizer of another flag of $\mc P$. Hence $\Gamma({\mc P}) \cong \mon{\mc P}$ (for further details of this fact see~\cite{MonPelWil11a}).

This leads to a useful reinterpretation of the condition for a regular polyhedron \mc P to be a cover of \mc Q. The fact that  \mc Q admits a flag action by $\Gamma(\mc P)$ is equivalent to there being an epimorphism from $\mon {\mc P}$ to $\mon {\mc Q}$. Thus,
we find it more natural to understand the cover $\mc P\searrow \mc Q$ as an epimorphism of monodromy groups, instead of as a homomorphism from an automorphism group to a monodromy group. This perspective is motivated by the natural way in which $i$-adjacent flags of $\mc P$ are mapped into $i$-adjacent flags of $\mc Q$. Henceforth we shall proceed according to this notion and use the generators $r_0, r_1, r_2$ of $\mon{\mc P}$ instead of those of $\Gamma({\mc P})$ to denote the action on the flags of $\mc Q$.

Several geometric objects can be viewed as abstract polyhedra, for example the convex polyhedra and the plane tilings. In such cases, the {\em symmetry group} is the group of isometries of the ambient space preserving the polyhedron. Every symmetry can be understood as an automorphism of the abstract object, but in general not every automorphism can be realized as a symmetry of a geometric realization of the abstract object. Throughout the remainder of this paper we will use the term ``polyhedra'' to mean either the geometric objects or abstract polyhedra, as appropriate.

An {\em Archimedean} or {\em uniform tiling} of the Euclidean plane is a non-regular tiling by regular convex polygons such that its automorphism group acts transitively on the vertices. There are eight Archimedean tilings, determined by the types and cyclic order of the regular polygons meeting at each vertex. Following ~\cite{GruShe87}, we denote each Archimedean tiling by a symbol of the form $(p_{1}.p_{2}.\ \ldots\ .p_{k})$ where each $p_{i}$ indicates a $p_{i}$-gon in a cyclic ordering of the faces around a vertex. Thus $(3.6.3.6)$ denotes the Archimedean tiling of the plane in which each vertex is surrounded by a triangle, a hexagon, a triangle and a hexagon (or any cyclic reordering or reversal of this sequence).

Every Archimedean tiling of the plane contains faces of different sizes.
In some of these tilings each flag orbit $\mathcal{O}$, under the symmetry group, is determined by the size of the faces in flags in $\mathcal{O}$ together with the size of the faces of their $2$-adjacent flags, thus giving a combinatorial description of the flag orbits.
More generally, each flag orbit $\mathcal{O}$, under the symmetry group of an Archimedean tiling of the plane, is determined by the sizes of the faces in flags in $\mathcal{O}$ together with the sizes of the faces in images of the flags in $\mathcal{O}$ by some monodromy elements which depend only on the tiling. This combinatorial description of the flag orbits under the symmetry group implies that the symmetry group and the automorphism group of any Archimedean tiling are isomorphic. Whenever $\mathcal{A}$ is an Archimedean tiling we can abuse notation and understand the elements of $\Gamma(\mathcal{A})$ either as combinatorial automorphisms or as symmetries of the geometric tessellation as needed. The eight Archimedean tilings are listed in Table \ref{t:wO}.

A {\em toroidal quotient} of a tiling $\mathcal{T}$ of $\mathbb{E}^2$ is a quotient $\mathcal{T}/\langle t_1, t_2 \rangle$ where $t_1$ and $t_2$ are two translations with respect to linearly independent vectors that are also symmetries of $\mathcal{T}$. The vertices, edges and faces of $\mathcal{T}/\langle t_1, t_2 \rangle$ are the orbits of the vertices, edges and faces of $\mathcal{T}$ under the action of $\langle t_1, t_2 \rangle$, and the incidence in $\mathcal{T}/\langle t_1, t_2 \rangle$ is the natural incidence induced from that in $\mathcal{T}$.

\section{Structure of the covers}\label{sec:structure}

In this section we show that the monodromy group of every Archimedean tiling $\mathcal{A}$ is a finite extension of a free abelian group of finite rank. As we shall see, the rank of such an abelian group is $2k$, where $k$ is the number of flag orbits of $\mathcal{A}$ under $\Gamma(\mathcal{A})$. Specific details of the monodromy groups of the Archimedean tilings are given in Lemmas \ref{l:w1} to \ref{l:quotientmap}. The main result is stated as Theorem \ref{t:representation}.

\begin{lemma}\label{l:w1}
Let $\mathcal{A}$ be an Archimedean tiling of $\EE^{2}$.  There exists an orbit $\mathcal{O}$ of flags under the action of $\Gamma(\mathcal{A})$, such that there exist $z_{\mathcal{O}}, w_{\mathcal{O}} \in \mon{A}$ fixing all flags of $\mathcal{A}$ not in $\mathcal{O}$ while acting on flags in $\mathcal{O}$ like translational symmetries of \mc A in two linearly independent directions.
\end{lemma}

\begin{proof}
In Table \ref{t:wO} we show an element $z_\mathcal{O}$ for each of the choices of $\mathcal{A}$. In each case, the orbit $\mathcal{O}$ is represented by the flag in gray in Figure \ref{f:archtil}. For each tessellation, the translation vector is with respect to the basis indicated in the same figure, where the white-head arrow indicates the first vector of the basis.
%\acom{GW}{Do you have a figure collection somewhere for this Daniel, or do we need to start drawing pictures?}\acom{DP}{I have the pictures, I'll put them in shortly (perhaps on Monday)}

%\acom{MM}{For this to make sense we need to have an ordered basis for each tessellation.  Right now we just have two basis vectors, but it is not clear which is the first}

%\acom{DP}{Done now, let me know if that is enough}

%\acom{MM}{That's a clever way of indicating things}

\begin{center}
\begin{table}[htb]
\centering
{\footnotesize
\begin{tabular}{|c|c|c|}
\hline
 \textbf{$\mathcal{A}$} & \textbf{$z_\mathcal{O}$} & \textbf{Translation vector} \\
 \hline
(3.6.3.6) & $[(r_0 r_1)^3 r_2 r_1 r_2]^2$ & $(2,0)$ \\
\hline
(4.8.8) & $[(r_0 r_1)^4 r_2]^4$ & $(4,0)$ \\
\hline
(3.12.12) & $[(r_0 r_1)^3 r_2]^3 [(r_1 r_0)^3 r_2]^3$ & $(2,0)$ \\
\hline
(3.4.6.4) & $(r_0r_1)^3r_2(r_0r_1)^6r_2(r_1r_0)^3r_2(r_0r_1)^6r_2$ & $(2,0)$ \\
\hline
(3.3.4.3.4) & $[(r_1r_0)^3r_2r_1r_2]^2[(r_0r_1)^3r_2r_1r_2]^2$ & $(1,1)$ \\
\hline
(3.3.3.4.4) & $[(r_0r_1)^3r_2(r_0r_1)^2r_2r_0r_1r_2(r_1r_0)^2r_2r_1r_2r_1r_2]^2$ & $(2,0)$ \\
\hline
(4.6.12) & $(r_0r_1)^4[r_2(r_0r_1)^{-4}]^2r_2(r_0r_1)^6r_2[(r_0r_1)^4r_2]^2(r_0r_1)^{-4}r_2(r_0r_1)^6r_2$ & $(2,2)$ \\
\hline
(3.3.3.3.6) & $[(r_0r_1)^3r_2r_1r_2r_0r_1r_2]^4$ & $(4,0)$\\
\hline
\end{tabular}
}
%\addtocounter{table}{-1}
\caption[]{\label{t:wO}Element $z_\mathcal{O} \in \mon{\mathcal{A}}$
}

\end{table}
\end{center}
%\gcom{I've gone through and checked all of these. I found (what I think) are a number of errors. For $(4.8.8)$ the conjugated version should be with $101$, not $10$ (and there's a typo there anyway). For $(3.12.12)$ the conjugated version should be with $0101$, not $10$. For $(3.4.6.4)$ there was a typo I fixed, but the given diagram doesn't match the directions of translation. For $(4.6.12)$ the conjugators should be $1010$, not $10$. Assuming this is what you intended, the direction for $w$ is wrong in the diagram. For $(3.6.3.6)$ the direction for $w$ appears to be incorrect in the diagram.}
%\acom{DP}{I corrected the elements of conjugation. You were right in all cases. Did you correct the typoes you mention? I don't see them. For all I see, the direction of the translation vector is fine in (3.6.3.6). I rotated the flag for 3.4.6.4 so that the direction matches with one of the arrows of the figure. For (4.6.12) I changed the arrows so that the one going up faced downwards.}
%\gcom{Once I get confirmation from you on the generators, I'll rerun the computations for the appendix. At this point, Mark and I are planning on putting most of that stuff on a digital archive at UAF once we see what section 4 looks like.}
%\acom{DP}{As far as I see, the generators are fine. If anything, we may discuss about the pictures.}
\begin{figure}[htb]
\begin{center}
\includegraphics[width=4in, height=4in]{v20i2p20/ArchTil1.pdf}
\caption{Bases for the translation subgroups of the symmetry groups of the Archimedean tilings. %\green{The labels here need to be modified to include parentheses.}\red{done}
}
\label{f:archtil}
\end{center}
\end{figure}

In all cases we can obtain $w_\mathcal{O}$ by conjugating $z_\mathcal{O}$ with an element in $\mon{A}$ fixing orbit $\mathcal{O}$. For instance
\begin{equation}\label{eq:wO}
\begin{array}{rcl}
(3.6.3.6) & \rightarrow & w_\mathcal{O} = r_1 r_0 z_\mathcal{O} r_0 r_1, \\
(4.8.8) & \rightarrow & w_\mathcal{O} = r_1 r_0 r_1 z_\mathcal{O} r_1 r_0 r_1, \\
(3.12.12) & \rightarrow &  w_\mathcal{O} = r_1 r_0 r_1 z_\mathcal{O} r_1 r_0 r_1,\\
(3.4.6.4) & \rightarrow & w_\mathcal{O} = r_0 r_1 z_\mathcal{O} r_1 r_0, \\
(3.3.4.3.4) & \rightarrow & w_\mathcal{O} = r_1 r_0 z_\mathcal{O} r_0 r_1, \\
(3.3.3.4.4) & \rightarrow & w_\mathcal{O} = r_2 r_1 r_0 r_1 r_2 z_\mathcal{O} r_2 r_1 r_0 r_1 r_2,\\
(4.6.12) & \rightarrow & w_\mathcal{O} = r_1 r_0 r_1 z_\mathcal{O} r_1 r_0 r_1, \\
(3.3.3.3.6) & \rightarrow & w_\mathcal{O} = r_1 r_0 z_\mathcal{O} r_0 r_1. \\
\end{array}
\end{equation}
\end{proof}


\begin{lemma}\label{l:wj}
Let $\mathcal{A}$ be an Archimedean tiling of $\EE^{2}$ and let $\mathcal{O}_1, \dots, \mathcal{O}_k$ be its flag orbits under $\Gamma(\mathcal{A})$. Then, for each $j \in \{1, \dots, k\}$ there exist $z_{j}, w_{j} \in \mon{A}$ fixing all flags of $\mathcal{A}$ not in $\mathcal{O}_j$ while acting on flags in $\mathcal{O}$ like translational symmetries of \mc A in two linearly independent directions.
\end{lemma}

\begin{proof}
Without loss of generality assume that $z_1$ and $w_1$ are respectively $z_{\mathcal{O}}$ and $w_{\mathcal{O}}$ as in Lemma \ref{l:w1}. Recall that the action of the monodromy group is transitive on the flags of $\mathcal{A}$. This implies that if a flag on orbit $\mathcal{O}_1$ is reached from a flag $\Phi$ on orbit $\mathcal{O}_j$ by the element $f \in \mon{A}$ then we may define $z_j$ and $w_j$ as $fz_1f^{-1}$ and $fw_1f^{-1}$ respectively. To see that $z_j$ and $w_j$ act
on flags in $\mathcal{O}_j$ like translational symmetries of \mc A in two linearly independent directions, it suffices to consider the corresponding action on the triangle of the barycentric subdivision of the tiling $\mc A$ corresponding to $\Phi$ and the triangle corresponding to $\Phi f$. %Since $\Phi f$ and $\Phi f z_{1}$ are in the same geometric orientation, so too will be $\Phi$ and $\Phi fz_{1}f^{-1}$.
Thus the geometric effect of $fz_{1}f^{-1}$ on $\Phi$ is the translation that maps $\Phi f$ to $\Phi f z_{1}$.
\end{proof}

In the following three lemmas we concentrate our attention on properties of the group $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$.

\begin{lemma}\label{l:commute}
Let $\mathcal{A}$, $z_{j}$, and $w_{j}$ be as in Lemma \ref{l:wj}. Then $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle \cong \mathbb{Z}^{2k}$.
\end{lemma}

\begin{proof}
By definition, an element in $\mon{A}$ equals $id$ if and only if it acts like $id$ on all flags of $\mathcal{A}$. Note that the elements $z_i z_j z_i^{-1} z_j^{-1}$, $w_iw_j w_i^{-1} w_j^{-1}$ and $z_{i}w_{j}z_{i}^{-1}w_{j}^{-1}$ act on all flags of $\mathcal{A}$ like a composition of two (maybe trivial) translational symmetries of \mc A and their inverses. Since the action of  any pair of translations commute, $z_i z_j z_i^{-1} z_j^{-1}$, $w_iw_j w_i^{-1} w_j^{-1}$ and $z_{i}w_{j}z_{i}^{-1}w_{j}^{-1}$ act like $id$ on all flags, implying that every pair of generators of $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ must commute. Clearly $z_i$ and $w_i$ have infinite order, and Lemma \ref{l:wj} implies that none of these generators can be expressed in terms of the others. Thus the lemma holds.
\end{proof}

\begin{lemma}\label{l:finiteindex}
Let $\mathcal{A}$, $z_{j}$, and $w_{j}$ be as in Lemma \ref{l:wj}. Then $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ has finite index in $\mon{A}$.
\end{lemma}

\begin{proof}


Let $t_1$ and $t_2$ be the translations in $\Gamma(\mc A)$ such that if $\Phi \in \mathcal{O}_1$ then $\Phi z_1 = t_1 \Phi$ and $\Phi w_1 = t_2 \Phi$. Note that, for any $j \in \{1, \dots, k\}$, there exists $\Phi \in \mathcal{O}_j$ such that $\Phi z_j = t_1 \Phi$ and $\Phi w_j = t_2 \Phi$. In general, any flag $\Phi \in \mathcal{O}_j$ satisfies that $\Phi z_j = \bar{t}_1 \Phi$ and $\Phi w_j = \bar{t}_2 \Phi$ for some translations $\bar{t}_1$ and $\bar{t}_2$ with respect to the images of the translation vectors of $t_1$ and $t_2$ under some symmetry of $\mathcal{A}$.

Let $M$ be the maximum of the lengths of the translation vectors $t_1$ and $t_2$. Then every pair of points of the parallelogram determined by $\bar{t}_1$ and $\bar{t}_2$ have a distance less than $2M$ between them.
We claim that, given any element $x \in \mon{A}$, there exists an element $y \in \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ such that, for any flag $\Phi$ of $\mathcal{A}$, the (Euclidean) distance between the vertices of $\mc A$ in $\Phi$ and $\Phi x y$ is less than $2M$. In fact, if for some flag orbit $j_0$ it happens that the distance between the vertices of $\Phi$ and $\Phi x$ is greater or equal than $2M$, then there exist appropriate powers $\alpha_1$ and $\alpha_2$ of the translations $\bar{t}_1$ and $\bar{t}_2$ defined above, such that the distance between the vertices of $\Phi$ and $\bar{t}_1^{\alpha_1} \bar{t}_2^{\alpha_2} \Phi x$ is less than $2M$ (it is enough to express in terms of $\bar{t}_1$ and $\bar{t}_2$ the translation mapping the vertex of $\Phi x$ to the parallelogram determined by the vertex of $\Phi$ together with the translations $\bar{t}_1$ and $\bar{t}_2$). By multiplying $x$ by the corresponding powers of $z_{j_0}$ and $w_{j_0}$ we obtain an element $x z_{j_0}^{\alpha_1} z_{j_0}^{\alpha_2}$ with the property that the distance between the vertices of $\Phi$ and $\Phi x z_{j_0}^{\alpha_1} z_{j_0}^{\alpha_2}$ is less than $2M$. This argument can be applied to all flag orbits.


Recall that any element of $\mon{A}$ is a permutation of the flags of $\mathcal{A}$. It follows from (\ref{eq:actions2}) that if $\Psi = \beta \Phi$ for some $\beta \in \Gamma(\mathcal{A})$ then the image of $\Psi$ under a given element $s \in \mon{A}$ is determined by $\beta$ and the image of $\Phi$ under $s$. This implies that to describe an element in $\mon{A}$ it suffices to define its action on only one flag of each flag orbit of $\mathcal{A}$ under $\Gamma(\mathcal{A})$.

Since there are finitely many flags whose vertices are at distance less than $2M$ from the vertex of $\Phi$, there are finitely many cosets of $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ in $\mon{A}$. Hence $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ has finite index in $\mon{A}$.

\end{proof}

\begin{lemma}\label{l:normal}
Let $\mathcal{A}$, $z_{j}$, and $w_{j}$ be as in Lemma \ref{l:wj}. Then the monodromy elements $z_j$ and $w_j$ can be chosen so that $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ is a normal subgroup of $\mon{A}$.
\end{lemma}

\begin{proof}
Let $Cl(z)$ denote the normal closure of $z$ in $\mon{A}$, and let $z_j$ and $w_j$ be as in the proofs of Lemmas \ref{l:w1} and \ref{l:wj}.  Since each $z_j$ and $w_j$ is a conjugate of $z_1$ it follows that $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle \le Cl(z_1)$.

On the other hand, every conjugate of $z_1$ acts like $id$ on all orbits of flags but one. Let $x \in \mon{A}$ and let $\mathcal{O}_{j}$ be the flag orbit containing flags where $x^{-1} z_1 x$ does not act like $id$. Then $x^{-1} z_1 x$ acts on any flag in $\mathcal{O}_j$ like a translation. From the actions of $x^{-1} z_1 x$, $z_j$, and $z_w$ on the flags of each Archimedean tessellation $\mathcal{A}$, it can be seen that $x^{-1} z_1 x$ is in the group $\langle z_j, w_j \rangle$. This shows that $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle \ge Cl(z_1)$ and the lemma holds.
%\acom{MM}{I changed the wording a tiny bit.  I don't like ``easy to see" when I had a hard time seeing it.}
\end{proof}

In what follows we derive a procedure to obtain a presentation in terms of generators and relations of $\mon{A}$, for any given Archimedean tiling $\mathcal{A}$. %\acom{GW}{Actually, I'm pretty sure this would work for any periodic tiling (perhaps with suitable additional criteria), but we'll have to see what we think after we are all done.}\noindent
We first define a finite map that arises from the quotient of $\mon{A}$ by the normal subgroup $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ as in Lemma \ref{l:normal}. The following lemma shows that such a map is in fact a toroidal quotient of $\mathcal{A}$.

\begin{lemma}\label{l:quotientmap}
Let $\mathcal{A}$ be an Archimedean tiling of $\EE^{2}$, let $z_j$, $w_j$ be as in Lemma \ref{l:wj}, chosen to satisfy the conditions in Lemma \ref{l:normal}, and let $t_1, t_2$ be the translations in $\Gamma(\mathcal{A})$ such that $t_1 \Phi = \Phi z_1$ and $t_2 \Phi = \Phi w_1$ for some flag $\Phi$.  Then $$\mon{A} / \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle \cong \mon{A/\langle t_1, t_2 \rangle}.$$
\end{lemma}%\acom{GW}{Might want to put the assume statement before the let.}\acom{DP}{I'm not sure of what you mean, but it sounds like some wording is necessary. I guess it is better if yo do it directly}
%\acom{GW}{This is what I meant. Picky logic stuff, but we can't retroactively pick the structure on something. We have to do it when we pick the element.}

\begin{proof}
Since there is a rank and adjacency preserving function from the flags of $\mathcal{A}$ to the flags $\mathcal{A}/\langle t_1, t_2 \rangle$ it follows that there is a surjective group homomorphism from $\mon{A}$ to $\mon{A/\langle t_1, t_2 \rangle}$ (see~\cite{MonPelWil11a} where these functions are called rap-maps). The kernel $H$ of such homomorphism contains $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ since all the monodromy elements in the latter preserve all flags in $\mathcal{A}/\langle t_1, t_2 \rangle$. On the other hand, the action of every element in $H$ equals the action of some product of generators $z_j$ and $w_j$ implying that $H = \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$.
\end{proof}

Note that the map $\mathcal{A}/\langle t_1, t_2 \rangle$ lies on the torus, and in particular, its monodromy group is finite. This allows us to compute a presentation of $\mon{A/\langle t_1, t_2 \rangle}$ in terms of generators and relations. Assume that $p$ is the least common multiple of the co-degrees of the faces, and $q$ is the degree of the vertices of $\mathcal{A}$, let $\pi$ be the natural quotient from $[p,q]$ to $\mon{A}$, and suppose that
$$\mon{A/\langle t_1, t_2 \rangle} = [p,q]/ Cl_{[p,q]}(\{g_1, \dots, g_m\})$$
for some $g_i \in [p,q]$, where $Cl_{\Lambda}(X)$ denotes the normal closure of $X$ in the group $\Lambda$. Then, for $i \in \{1, \dots, m\}$ there exists $h_i \in \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$, such that

\begin{equation}\label{eq:liftquotient}
\pi(g_i) = h_i.
\end{equation}%\acom{GW}{This doesn't make any sense. $g_{i}$ in $[p,q]$, but $\pi:[p,q]\to\text{Mon}(\mc A)$ and $h_{i}\in \text{ker}(\pi)$. I'm not sure how to fix what you mean here. And I'm not sure the next sentence helps me resolve this confusion.}
%\acom{DP}{Does it make sense now?}\gcom{All good.}
Here we are assuming that $h_i$ is presented as an expression in terms of $\{z_j, w_j \mid j \in \{1, \dots, k\}\}$, whereas $\pi(g_i)=\tilde g_{i}$ has the same expression in terms of the generators $r_0, r_1, r_2$ of $\mon{A}$ as $g_i$ does in terms of the generators $\rho_0, \rho_1, \rho_2$ of $[p,q]$.

It follows from Lemma \ref{l:commute} that $\mon{A}$ satisfies the relations
\begin{equation}\label{eq:commute}
  \begin{array}{ccc}
    z_i z_j & = & z_j z_i,\\
    w_iw_j & = & w_j w_i,\\
    z_i w_j & = & w_j z_i,
  \end{array}
\end{equation}
for $i, j \in \{1, \dots, k\}$.

From Lemma \ref{l:normal} we conclude that for $i \in \{0,1,2\}$ and for $j \in \{1, \dots, k\}$ there exist a word $x_{i,j}$ and a word $y_{i,j}$ on the set $\{z_l, w_l\mid l \in \{1, \dots, k\}\}$ such that
\begin{equation}\label{eq:normalize}
  \begin{array}{ccc}
   r_i z_j r_i & = & x_{i,j}\\
   r_i w_j r_i & = & y_{i,j}
  \end{array}
\end{equation}

Clearly $\mon{A}$ satisfies (\ref{eq:liftquotient}), (\ref{eq:commute}) and (\ref{eq:normalize}). The next theorem shows that these relations suffice to describe $\mon{A}$ as a quotient of the Coxeter group $[p,q]$.

\begin{theorem}\label{t:representation}
Let $\mathcal{A}$ be an Archimedean tiling of $\EE^{2}$ with vertices of degree $q$ such that the least common multiple of the co-degrees of its faces is $p$. Let $z_j$ and $w_j$ be as in Lemma \ref{l:normal}. Then $\mon{A}$ is the quotient of the Coxeter group $[p,q]$ by the extra relations given in (\ref{eq:liftquotient}), (\ref{eq:commute}) and (\ref{eq:normalize}), where $r_i$ in (\ref{eq:normalize}) should be interpreted as the corresponding generator $\rho_i$ of $[p,q]$.
\end{theorem}

\begin{proof}
%{\bf To be done by Gordon}
This is equivalent to proving that the relations  on the $r_{i}$ given by (\ref{eq:liftquotient}), (\ref{eq:commute}) and (\ref{eq:normalize}) form a sufficient set of relations to determine if two
 arbitrary words in $r_{0}, r_{1}$ and $r_{2}$ represent the same word in $\mon{A}$.

Let $g_{\omega}\in\mon A$ be represented by a word $\omega$ on the $r_{i}$, and let $ \omega_{i}$ be representatives from each of the cosets of $\cl{[p,q]}{\{\tilde g_1, \dots, \tilde g_m\}}$ in $\mon A$.
Lemma \ref{l:normal} implies that we can rewrite $g_{\omega}=w_{i}\beta$, where $\beta\in \cl{[p,q]}{\{\tilde g_1,  \dots, \tilde g_m\}}$. Thus $\beta=\beta_{1}\tilde g_{i_{1}}\beta_{1}^{-1}\beta_{2}\tilde g_{i_{2}}\beta_{2}^{-1}\cdots\beta_{n}\tilde g_{i_{n}}\beta_{n}^{-1}$ for some choice of $\beta_{i}$ words on the $r_{i}$ and $\tilde g_{i_{k}}\in\{\tilde g_1,  \dots, \tilde g_m\}$.
 By (\ref{eq:liftquotient}) $$g_{\omega}=\omega_{i}\beta_{1} h_{i_{1}}\beta_{1}^{-1}\beta_{2} h_{i_{2}}\beta_{2}^{-1}\cdots\beta_{n} h_{i_{n}}\beta_{n}^{-1}$$ where $h_{i_{j}}\in\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle.$ By (\ref{eq:normalize}),  $z_{j}r_{i}=r_{i}x_{ij}$ and $w_{j}r_{i}=r_{i}y_{ij}$.
  Since $x_{ij}, y_{ij}\in\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$, repeated applications of these relations show that $\beta_{k}h_{k_{l}}\beta_{k}=\beta_{k}\beta_{k}^{-1}\hat h_{k_{l}} = \hat h_{k_{l}}$, where $\hat h_{k_{l}}\in \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$. Thus $g_{\omega}=\omega_{i}\Pi_{i=1}^{n}\hat h_{i_{j}}=\omega_{i}t$ for some $t\in \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$.
   By the commutativity of the generators of $\langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle$ obtained as a consequence of (\ref{eq:commute}), $t$ may be represented uniquely as a product of the form $\Pi_{i=1}^{k}(z_{i}^{p_{i}}w_{i}^{q_{i}})$ for some choice of $p_{i},q_{i}\in\ZZ$. Thus every element of $\mon A$ has a unique presentation of the form $\omega_{i}\Pi_{j=1}^{k}(z_{j}^{p_{j}}w_{j}^{q_{j}})$, completing the argument.
\end{proof}
%\gcom{Let me know if you think I've got it with this presentation of the argument. We may need to include some additional commentary about Lemma 3.1 from that other paper of yours Daniel. Also, it seems to me that set up this way, we can subsume the relations for $[p,q]$ into the $g_{i}$ (alternately  $\tilde g{i}$) as relations on $[\infty,\infty]$. Not sure if that is useful, or if I need to make more explicit use of these Coxeter relations in the argument.}
%\acom{DP}{I agree with the presentation of the argument modulo some wording and notation (for instance, $\tilde{h}_i$ was never defined and I guess they should be $h_i$'s). Perhaps it would be a good idea to put some emphasis (few words) on why the presentation is unique. Feel free to add any lemma that would contribute for clarity of the proof of the theorem. It is true that the theorem would still be true from the viewpoint of $[\infty, \infty]$, but I don't know if it is better to state in such generality. When we use $[p,q]$ we are already assuming that the reader understands that there is a flag-action of $[p,q]$ on $\mathcal{A}$, but other than that, I feel that it is nicer to have this intermediate (``smaller'') cover than the universal one.}
%\gcom{That's right on the $h_{i}$s, they were the result of a copy and paste error. I'm confused about the uniqueness you refer to. In general, presentations for groups are far from unique. Perhaps I misunderstand?}\acom{DP}{Sorry, I meant the uniqueness of the presentation of each element as $\omega_{i}\Pi_{j=1}^{k}(z_{j}^{p_{j}}w_{j}^{q_{j}})$.}

\begin{remark}
\label{rem:minimal}
To conclude this section we note that the translation vectors in Table \ref{t:wO} were verified to be of minimal length possible. To do this we analyzed, with the help of GAP~\cite{GAP}, the action of $\mon{A}$ on the toroidal quotient of $\mathcal{A}$ determined by the translation vectors of $z_{\mathcal{O}}$ and $w_{\mathcal{O}}$, and noticed that no element in $\mon{A}$ acts like a translation on flags with preimage in $\mathcal{O}$ while fixing the others.
\end{remark}

\section{The Minimal Regular Cover of (3.4.6.4)}\label{sec:3464}

%{\bf Mark writes}
In this section we demonstrate how the theoretical results of the previous section are used to compute the monodromy group of the $(3.4.6.4)$ tiling $\mathcal{A}$.  The group $\Gamma (\mathcal{A} )$ acts with four orbits on the flags of $\mathcal{A}$, and thus, following Lemma~\ref{l:wj}, we define $z_j$ and $w_j$ for $j \in \{1,\ldots,4\}$.  To describe the action of these monodromy elements in terms of translations, we pick a basis $(b_1,b_2)$ (see Figure~\ref{3464flags} for a representation of the four flag orbits and this basis).

\begin{figure}[h]
\begin{center}\includegraphics[width=2in]{v20i2p20/3464labeledflags2.pdf}\end{center}
\caption{The four flag orbits and a basis for the translation subgroup of the symmetry group of $(3.4.6.4)$.}\label{3464flags}
\end{figure}%\gcom{I will make a nicer version of this graphic before we go to press.}

Following Table~\ref{t:wO} and Equation~(\ref{eq:wO}) we define:
$$z_1 := (r_0r_1)^3r_2(r_0r_1)^6r_2(r_1r_0)^3r_2(r_0r_1)^6r_2 \textrm{ and } w_1 := r_0r_1 z_1 r_1 r_0.$$

\noindent Then, the proof of Lemma~\ref{l:wj} shows how to construct the remaining $z_j$ and $w_j$.  This yields:
\begin{center}
\begin{tabular}{ll}
$z_2 = r_2 z_1 r_2   $ &$   w_2 = r_2 w_1 r_2   $ \\
$z_3 = r_1 r_2 z_1 r_2 r_1    $& $  w_3 = r_1 r_2 w_1 r_2 r_1  $  \\
$z_4 = r_2 r_1 r_2 z_1 r_2 r_1 r_2   $ &$   w_4 =  r_2 r_1 r_2 w_1 r_2 r_1 r_2    $ \\
\end{tabular}
\end{center}
Let $\Phi$ be the shaded flag in Figure~\ref{3464flags} that is in orbit 1.  The monodromy elements $z_1$ and $w_1$ act trivially on any flag not in the same orbit of $\Phi$ and act like  translation on any flag in orbit 1.  In particular, if $t_1= (2,0)$ and $t_2 = (0,2)$; then $t_1 \Phi = \Phi z_1$ and $t_2 \Phi = \Phi w_1$.  Furthermore, each $z_j$ and $w_j$ is a conjugate of $z_1$, and thus the conditions in Lemma~\ref{l:normal} are satisfied.   Therefore, by Lemma~\ref{l:quotientmap},

$$\mon{A} / \langle z_j, w_j \mid j \in \{1, \dots, k\} \rangle \cong \mon{A/\langle t_1, t_2 \rangle}.$$


In this case, the quotient $\mathcal{A}/\langle t_1, t_2 \rangle$ consists of four hexagons, twelve squares, and eight triangles on the torus, and therefore has 192 flags.  Thus, the group $\mon{A/\langle t_1, t_2 \rangle}$ can be constructed as a subgroup of the symmetric group $S_{192}$.  Then, a presentation for the group can be computed in GAP.
%\acom{MM}{Where exactly will we put the code?}
%\gcom{I will talk to our library staff about how that all works after I get back from Mexico.}
%\acom{DP}{Do we need to put the code? I would just say that we computed it. Anyway, I don't object to putting it online if someone finds out how and where.}
%\acom{MM}{I guess we don't need to put the code online, but I made a ``well" documented version of the program in case the referee or someone in the future wants to see how it works.}

The result of this calculation is that $\mon{A/\langle t_1, t_2 \rangle}$ is the quotient of the Coxeter group $[12,4]$ by the relators:

$$g_1= (\rho_2(\rho_0\rho_1)^4 \rho_2\rho_1(\rho_0\rho_1)^3)^2, $$
$$g_2 = \rho_2(\rho_0\rho_1)^5\rho_2\rho_1\rho_2(\rho_0\rho_1)^3\rho_2(\rho_0\rho_1)^2\rho_2(\rho_0\rho_1)^4\rho_2\rho_1,$$
$$g_3=(\rho_1\rho_2(\rho_1\rho_0)^3\rho_2(\rho_1\rho_0)^5)^2.$$

Following Theorem~\ref{t:representation}, the monodromy group of $\mathcal{A}$ is the quotient of the Coxeter group $[12,4]$ by extra relations of types~(\ref{eq:liftquotient}), (\ref{eq:commute}) and (\ref{eq:normalize}).  To determine the $h_i$ for the relations of type (\ref{eq:liftquotient}), we must determine the action of $g_i$ on the flags of $\mathcal{A}$.  The result of this computation is that:
$$g_1 = Id,$$
$$g_2 = z_3^{-1} w_3,$$
$$g_3 = z_3.$$

The twenty eight relations of type~(\ref{eq:commute}) are trivial to write down, and all that remains to find a complete presentation for the monodromy group is to calculate the $x_{i,j}$ and $y_{i,j}$ for the relations of type~(\ref{eq:normalize}).  This is straight forward, and the result is shown in the following table.  Each entry in the table is the result of conjugating the element of a column label by the element of the row label.

%\acom{MM}{What do we think about this format to use in the appendix for the other tilings?}
%\gcom{Certainly makes things tidier and more compact. May need to transpose it for some of the tilings to get it to fit.}
%\acom{DP}{I agree that it looks good. Perhaps we can make the font smaller if it doesn't fit for other tilings.}

\begin{table}[htb]\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|} \hline
 & $z_1$  &$w_1$  & $z_2 $ & $w_2$ &$z_3$ & $w_3 $ &$z_4 $ &$w_4$ \\ \hline \hline
 $r_0$ & $z_1$  & $w_1^{-1} z_1$ & $z_2 $& $w_2^{-1} z_2 $&
  $z_3^{-1}$& $z_3^{-1} w_3$ & $z_4^{-1}$  &  $w_4 z_4^{-1}$  \\ \hline
 $r_1$  & $z_1 w_1^{-1}$  & $w_1^{-1}$ & $z_3$ & $w_3$ & $z_2$ & $w_2$ & $w_4^{-1}z_4$  & $w_4^{-1}$  \\ \hline
$r_2$  & $z_2$  &  $w_2$& $z_1$ &  $w_1$ & $z_4$ & $w_4$  &  $z_3$ & $w_3$  \\ \hline
\end{tabular}\end{center}
\caption{The monodromy element in the table is equal to the element obtained by conjugating the column label by the row label.}
\end{table}


The monodromy group of the $(3.4.6.4)$ tiling is obtained from the Coxeter group $[12,4]$ by adding the 3 relations of type~(\ref{eq:liftquotient}), the 28 relations of type~(\ref{eq:commute}), and the 24 relations of type~(\ref{eq:normalize}).    However, many of these relations are redundant. Using GAP, we were able to reduce this list of 55 extra relations down to a set of 20.  Our final presentation for $\mon{A}$ can be found in Appendix~\ref{Append3464}.  We include descriptions of the sets of necessary relations for the remaining tilings as Appendix~\ref{AppendB}.  Unlike for the $(3.4.6.4)$ tiling, the relations of type (\ref{eq:liftquotient}) are generally quite long and are instead found online at http://hdl.handle.net/11122/1232.


\section{Covers of the Archimedean Toroids}\label{sec:torus}
All vertex-transitive maps on the torus are toroidal quotients of regular or Archimedean tessellations of $\mathbb{E}^2$ ~\cite{PelWei11}, but not all such quotients are vertex-transitive. The minimal regular covers of (not necessarily vertex-transitive) toroidal maps obtained as toroidal quotients of the regular tessellations of $\mathbb{E}^2$ were found in~\cite{DraMix13}. %\acom{DP}{It turns out that Mark is a co-author. He should tell us the proper name. I understand that it hasn't been accepted yet.}
%\acom{MM}{Minimal covers of equivelar toroidal maps:
%Kostiantyn Drach, Mark Mixer  (Maxym dropped out)}
%\gcom{Are there more details?}
%\acom{DP}{This should be the paper that you and Kostya submitted, not the one in preparation. There are two different papers, right?}
All such covers are toroidal maps with Schlafli type $\{3,6\}$, $\{4,4\}$, or $\{6,3\}$. In this section we discuss an approach to finding the minimal regular covers of toroidal quotients of the Archimedean tilings using the minimal regular covers of these tilings. For some of the toroidal quotients we explicitly describe how to obtain the minimal regular cover.

Let $\mathcal{A}$ be an Archimedean tiling. The elements $z_{\mathcal{O}}$ as in Table \ref{t:wO} and $w_{\mathcal{O}}$ as in (\ref{eq:wO}) in $\mon{A}$ act on flags in all but one flag orbit (namely $\mathcal{O}$) as $id$, whereas they act on a given flag $\Phi \in \mathcal{O}$ like translations $t_1$ and $t_2$. Moreover, the translation vectors in Table \ref{t:wO} were chosen so that their norm is minimal possible (see Remark \ref{rem:minimal}).
It is easy to see, then, that if $x \in \mon{A}$ acts on all flags not in $\mathcal{O}$ like $id$ and acts on $\Phi$ like a translation $t_3$ then $t_3 \in \langle t_1, t_2 \rangle$.

Lemmas \ref{l:commute}, \ref{l:finiteindex}, \ref{l:normal}, and \ref{l:quotientmap} hold if we substitute $z_{\mathcal{O}}$ by $z_{\mathcal{O}}^s$ for some integer $s \ge 2$ while defining $w_{\mathcal{O}}$ as in (\ref{eq:wO}) and $z_i$, $w_i$ as in the proof of Lemma \ref{l:wj}. In other words, given a normal finite index subgroup $H$ of $\mon{A}$ isomorphic to $\mathbb{Z}^{2k} = \langle y_1, \dots, y_{2k} \rangle$ as in Section \ref{sec:structure}, the subgroup of $H$ isomorphic to $\langle sy_1, \dots, sy_{2k} \rangle$ is also normal and has finite index. Moreover,
\begin{equation}\label{eq:toroidal}
\mon{A}/\langle z_j^s, w_j^s \mid j \in \{1, \dots, k\} \rangle \cong \mon{A / \langle t_1^s, t_2^s \rangle}
\end{equation}
if $z_j$ and $w_j$ are as in Lemma \ref{l:quotientmap}. The next theorem follows from (\ref{eq:toroidal}).

\begin{theorem}\label{t:toroidal}
Let $\mathcal{A}$ be an Archimedean tiling of $\EE^{2}$. Let $z_\mathcal{O}$ and $w_\mathcal{O}$ be as in Table \ref{t:wO} and (\ref{eq:wO}) respectively, let $t_1$ and $t_2$ be translations such that $t_1 \Phi = \Phi z_{\mathcal{O}}$ and $t_2 \Phi = \Phi w_{\mathcal{O}}$ for some flag $\Phi$, and let $z_i$ and $w_i$ be as in the proof of Lemma \ref{l:wj}. Let $\mathcal{A}_{t_1^s, t_2^s}$ be the toroidal map obtained from $\mathcal{A}$ by taking quotient by the translation group $\langle t_1^s, t_2^s \rangle$. Then $\mon{A_{t_1^s, t_2^s}}$ is obtained from $\mon{A}$ by imposing the single extra relation $z_{\mathcal{O}}^s = id$.
\end{theorem}

As an example, consider the map $\mathcal{M} = \mathcal{A}_{t_1^2, t_2^2}$ where $\mathcal{A}$ is the tiling $3.4.6.4$. Then, by Theorem \ref{t:toroidal} we know that the minimal regular cover of $\mathcal{M}$ is obtained as the quotient of the Coxeter group $[12,4]$ by the elements in the table of Appendix~\ref{Append3464} plus the single extra relation $z_1^2 = id$.%, which is equivalent to
%\[(\rho_0\rho_1)^3\rho_2(\rho_0\rho_1)^6\rho_2(\rho_1\rho_0)^3\rho_2(\rho_1\rho_0)^6\rho_2 = \rho_2(\rho_0\rho_1)^6\rho_2(\rho_0\rho_1)^3\rho_2(\rho_1\rho_0)^6\rho_2(\rho_1\rho_0)^3.\]

%\begin{remark}
We note that if $\mathcal{A}_{t_1, t_2}$ is as in Theorem \ref{t:toroidal} then Relations (\ref{eq:commute}) and (\ref{eq:normalize}) become redundant in $\mon{A_{t_1, t_2}}$ because of the extra relation $z_{\mathcal{O}} = id$. This is consistent with what we knew, since the only defining relations of $\mathcal{A}_{t_1, t_2}$ are (\ref{eq:liftquotient}).%This implies that $\mon{\mathcal{A}_{t_1, t_2}}$ can be described with as many relators as those in (\ref{eq:liftquotient}) plus one in terms of the generators $r_0$, $r_1$, $r_2$.
%\end{remark}

So far we have described the minimal regular covers of toroidal maps obtained as quotients of Archimedean tessellations by the specific translation groups $\langle t_1^s, t_2^s \rangle$. These particular quotients cover any other toroidal quotient of the Archimedean tessellations
\cite{DraSkor13}.
%\acom{MM}{There paper is currently titled: Archimedean toroids and their minimal almost regular covers.}
 Consequently, the minimal regular cover of a toroidal quotient $\mathcal{A}_{t_1^{s_1} t_2^{r_1}, t_1^{s_2} t_2^{r_2}}$ of an Archimedean tiling is a quotient of the minimal regular cover of one of the toroidal maps $\mathcal{A}_{t_1^s, t_2^s}$, and, given any particular such tiling, its cover can be found easily by computing the stabilizer of a flag of $\mathcal{A}_{t_1^{s_1} t_2^{r_1}, t_1^{s_2} t_2^{r_2}}$ under the action of the monodromy group of its cover. The determination of these convenient covers $\mathcal{A}_{t_1^s, t_2^s}$ for arbitrary toroidal quotients of Archimedean maps is beyond the scope of this paper, hence we do not present a full description of them.

\section{Conclusions and Open Questions}


For each of the tilings described in this paper we have been able to identify a corresponding abelian extension of a finite group to which a minimal regular cover of the tiling may be associated. Critical to this construction are the existence of elements in the monodromy group of the tiling that act like translations in two different directions on one transitivity class of flags and fix all other classes. This is suggestive.
\begin{conjecture}\label{c:pairsExist} For each periodic tiling $\mc T$ of the plane there exist a pair of elements $w_{\mc O}, z_{\mc O}\in\mon{ T}$ with the property that $w_{\mc O},z_{\mc O}$ act like translations in two different directions on all flags in one transitivity class of flags under the symmetry group of $\mc T$, and fix all flags in the other transitivity classes.
\end{conjecture}

If the previous conjecture is true, the following conjecture would be a simple corollary by arguments similar to those presented in this paper.
\begin{conjecture} Every periodic tiling of the plane admits a representation of the form given in Theorem \ref{t:representation}.\end{conjecture}

 A broader question in the context of these results is the formulation of what it means for an infinite abstract polytope to be periodic in the absence of a geometric presentation (abstract polytopes are a generalization to higher ranks of abstract polyhedra, for more details see \cite{McMSch02}). A first candidate would be to consider those polytopes of rank $d$ whose automorphism groups contain a free abelian subgroup of finite index of rank $(d-1)$, or {\em translational periodic} polytopes;  in the case of the periodic tilings of the Euclidean plane the free abelian group is actually the group of translational symmetries of the tiling.

Given the context of the current work, a second candidate would be to consider those polytopes whose monodromy groups contain a free abelian subgroup of finite index of rank $(d-1)k$ where $d$ is the dimension (rank) of the abstract polytope, $k$ is the number of flag orbits under the automorphism group. This class of polytopes could be referred to as {\em monodromy periodic}.  A natural additional requirement on the $(d-1)k$ generators is that, for each flag orbit $\mathcal{O}$, $d$ generators act nontrivially on flags in $\mathcal{O}$ while fixing all flags in the other orbits. Is every translational periodic polytope automatically a monodromy periodic polytope? Conversely, is every monodromy periodic polytope a translational periodic polytope?  A contrasting approach to periodicity, in the context of aperiodic tilings, can be found in \cite{Sch03}, where a tiling of a Euclidean space by convex tiles is periodic if its
automorphism group contains an element of infinite order. 

The classification of automorphism groups of polytopes satisfying either of these definitions would be interesting since they would be freed of the geometric constraints of the crystallographic restriction, and so one might expect there to be more than the familiar 17 plane crystallographic groups, or the 230 crystallographic space groups. A related question about monodromy groups of abstract polytopes subject to the same restrictions is whether all such monodromy groups arise from Euclidean periodic tilings.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% \bibliographystyle{plain}
% \bibliography{myBibFile}
% If you use BibTeX to create a bibliography


\newpage
\begin{appendices}
\section{Presentation for the monodromy group of (3.4.6.4)}
\label{Append3464}
The monodromy group of the (3.4.6.4) tiling can be presented as the quotient of the Coxeter group $[12,4]$ by the following twenty additional relators.


\begin{table}[htb]
\centering
\begin{small}
\resizebox{.9\textwidth}{!}{
\begin{tabular}{|p{18cm}|}
\hline
$
(\rho_2 (\rho_1 \rho_0)^3 \rho_1 \rho_2 (\rho_1 \rho_0)^4))^2
$
\\ \hline
$
(\rho_2 (\rho_0 \rho_1)^5)^4
$
\\ \hline
$
(\rho_0 \rho_1)^5 \rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_0 \rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^6 \rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_0 \rho_2 \rho_1 \rho_0 \rho_2 \rho_1
$
\\ \hline
$
\rho_2 (\rho_0 \rho_1)^2 \rho_0 \rho_2 \rho_1 \rho_2 \rho_1 (\rho_0 \rho_1)^2 \rho_2 \rho_0 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_2 \rho_1 (\rho_0 \rho_1)^4 \rho_0 \rho_2 \rho_1 \rho_2 \rho_1 (\rho_0 \rho_1)^4 \rho_2 \rho_1
$
\\ \hline
$
(\rho_0 \rho_1)^3 (\rho_2 \rho_1 \rho_0)^3 (\rho_1 \rho_0)^3 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^4 (\rho_2 \rho_1 \rho_0 \rho_1)^2 \rho_0 \rho_1 \rho_0 \rho_2 (\rho_1 \rho_0)^3 \rho_2 \rho_1 \rho_2
$
\\ \hline
$
(\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 (\rho_2 \rho_0 \rho_1)^2 (\rho_0 \rho_1)^3 \rho_2 \rho_1 \rho_0 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_0 \rho_2 \rho_1 \rho_2 \rho_1 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0 \rho_1)^2  \rho_2 (\rho_0 \rho_1)^4 \rho_2
$
\\ \hline
$
(\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^3 (\rho_0 \rho_2 \rho_1)^2 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2
 \rho_2$\\
$ %2nd line
 (\rho_1 \rho_0)^2 \rho_2 \rho_1 \rho_2
$
\\ \hline
$
(\rho_0 \rho_1)^3 \rho_2 (\rho_1 \rho_0)^2 (\rho_1 \rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^3 \rho_1 \rho_2 \rho_0 (\rho_1 \rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2 \rho_0 \rho_1 \rho_2 (\rho_1 \rho_0)^4 \rho_1 \rho_2 (\rho_1 \rho_0)^2 \rho_1 \rho_2
$
\\
$ %2nd line
(\rho_0 \rho_1)^2 \rho_2 \rho_0 \rho_1 \rho_2
$
\\ \hline
$
\rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^3 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^4 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^2 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^5 \rho_2 (\rho_1 \rho_0)^2 \rho_1 \rho_2 (\rho_0 \rho_1)^2 \rho_2 $
\\
$ % 2nd line
(\rho_1 \rho_0)^3
 \rho_1 \rho_2 (\rho_1 \rho_0)^2 \rho_2 \rho_1
$
\\  \hline
$
\rho_1 \rho_0 \rho_1 \rho_2 (\rho_1 \rho_0)^4 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^3 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^5 (\rho_1 \rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2 (\rho_2 (\rho_1 \rho_0)^2)^2 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 \rho_2
$
\\
$ %2nd line
(\rho_1 \rho_0)^3\rho_2 \rho_1 \rho_0 \rho_1 \rho_2 \rho_1 \rho_0
$
\\ \hline
$
(\rho_1 \rho_0)^3 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^3 (\rho_2 \rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^5 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 \rho_0  \rho_1 \rho_2 $
\\
$ %2nd line
(\rho_1 \rho_0)^3 \rho_1 \rho_2
 (\rho_1 \rho_0)^2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 (\rho_2 \rho_0 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2
$
\\ \hline
$
\rho_0 \rho_1 \rho_2 (\rho_1 \rho_0)^4 \rho_1 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^3 (\rho_2 \rho_1 \rho_0 \rho_1)^2 (\rho_0 \rho_1)^4 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^2 \rho_1 \rho_2 (\rho_0 \rho_1)^2 \rho_0 (\rho_2 \rho_1)^2  (\rho_0 \rho_1 \rho_2)^2
$
\\
$ % 2nd line
(\rho_1 \rho_0)^5(\rho_2 \rho_1)^2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 (\rho_2 \rho_0 \rho_1)^2 \rho_2
$
\\ \hline
$
(\rho_0 \rho_2 \rho_1)^2 (\rho_0 \rho_1)^4 \rho_2 (\rho_1 \rho_0)^2 (\rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0 \rho_2)^2 (\rho_1 \rho_0)^3 \rho_2 (\rho_1 \rho_0)^2 (\rho_1 \rho_2 \rho_1 \rho_0)^2 \rho_2
$
\\
$ %2nd line
  (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^2 \rho_0 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^3 \rho_2 (\rho_1 \rho_0)^2 \rho_1
$
\\ \hline
$
(\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^5 (\rho_2 \rho_0 \rho_1)^2 (\rho_0 \rho_1)^4 \rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2  ((\rho_0 \rho_1)^3 \rho_2)^2 (\rho_1 \rho_0)^2 \rho_1 \rho_2
$
\\
$ % 2nd line
  (\rho_0 \rho_1)^3 \rho_2 (\rho_1 \rho_0)^4 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_0 \rho_2 \rho_1 \rho_2
$
\\ \hline
$
\rho_2 (\rho_1 \rho_0)^3 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_2 (\rho_1 \rho_0)^2)^2 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^5 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^2 (\rho_0 \rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^2  (\rho_0 \rho_2 \rho_1)^2 $
\\
$ %2nd line
(\rho_0 \rho_1)^2
\rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_1 \rho_0)^4 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^3 \rho_2 (\rho_1 \rho_0)^2
$
\\ \hline
$
\rho_0 \rho_1 (\rho_0 \rho_2 \rho_1)^2 \rho_0 \rho_1 \rho_0 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2 \rho_2
$
\\
$ %2nd line
(\rho_1 \rho_0)^3 \rho_2 (\rho_1 \rho_0)^4 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^3 \rho_2 \rho_1
$
\\ \hline
$
\rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^4 (\rho_2 (\rho_1 \rho_0)^3)^2 (\rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^4 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_1 \rho_0)^4 (\rho_1 \rho_2 (\rho_1 \rho_0)^2)^2 \rho_2 $
\\
$ %2nd line
(\rho_1 \rho_0)^3
 \rho_1 \rho_2 (\rho_0 \rho_1)^2 (\rho_2 \rho_0 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^2 \rho_2 \rho_1 \rho_0 \rho_1
$
\\ \hline
$
\rho_0 (\rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_2 (\rho_0 \rho_1)^4 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^3 (\rho_2 \rho_1 \rho_0)^2 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^4 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^4
$
\\
$ %2nd line
 (\rho_2 \rho_1 \rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^3 (\rho_0 \rho_2 \rho_1)^2 \rho_0 (\rho_1 \rho_0 \rho_2)^2 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^3
$
\\ \hline
$
\rho_0 \rho_1 \rho_0 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^3 \rho_1 \rho_0 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^2 \rho_1 \rho_2$
\\
$ %2nd line
 (\rho_1 \rho_0)^3
 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 \rho_2 \rho_1 \rho_0 (\rho_2 \rho_1)^2 (\rho_0 \rho_1)^3 \rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^3 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^4 \rho_2 \rho_1 \rho_2 \rho_0 \rho_1 \rho_2
$
\\ \hline
$
\rho_0 \rho_1 \rho_0 \rho_2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_2 (\rho_0 \rho_1)^3)^2 \rho_2 (\rho_0 \rho_1)^2 \rho_2 (\rho_0 \rho_1)^4 (\rho_2 \rho_1 \rho_0)^4 (\rho_1 \rho_0)^3 (\rho_1 \rho_2)^2 $
\\
$ %two lines
(\rho_0 \rho_1)^3
(\rho_2 (\rho_0 \rho_1)^2)^2 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_2)^2 (\rho_0 \rho_1)^5 (\rho_2 \rho_1 \rho_0)^2 (\rho_1 \rho_0)^2 \rho_2 (\rho_1 \rho_0)^2 \rho_2 \rho_1 \rho_0 \rho_2 \rho_1 \rho_2
$
\\ \hline
\end{tabular}}
\caption{Non-Coxeter relators for the presentation of $\mon{(3.4.6.4)}$}
\end{small}
\end{table}




%\newpage

\section{The remaining minimal regular covers}
\label{AppendB}
In this section we give a summary of the relations in the presentations of the monodromy groups for the (3.3.4.3.4), (3.3.3.4.4), (4.6.12), and (3.3.3.3.6) tilings.  We give these relations in terms of the monodromy elements $z_j$ and $w_j$ described in Lemma~\ref{l:wj}.  Following Theorem~\ref{t:representation}, the equations (\ref{eq:liftquotient}), (\ref{eq:commute}), and (\ref{eq:normalize}) define three different types of relations in the presentation.  The relations of type (\ref{eq:liftquotient}) will  be omitted here, as they are quite long.  However, the full list of these relations can be found online at http://hdl.handle.net/11122/1232.
%\gcom{I have uploaded the files to the data store. }
Those of type (\ref{eq:commute}) are trivial to write down and are omitted.  In each of the tables below we summarize the relations of type (\ref{eq:normalize}), each monodromy element in the table is equal to the element obtained by conjugating the column label by the row label.

\subsection{(3.3.4.3.4)}
\begin{center}
\begin{tabular}{ll}
$z_1=[(r_1r_0)^3r_2r_1r_2]^2[(r_0r_1)^3r_2r_1r_2]^2$   \hspace{2cm}  & $w_1= r_1 r_0 z_1 r_0 r_1$\\
$z_2 = r_0 z_1 r_0   $ &$   w_2 = r_0 w_1 r_0   $ \\
$z_3 = r_2 z_1 r_2   $ &$   w_3 = r_2 w_1 r_2   $ \\
$z_4 = r_2 r_0 z_1 r_0 r_2   $ &$   w_4 =  r_2 r_0 w_1 r_0 r_2    $ \\
$z_5 = r_1 r_2 r_0 z_1 r_0 r_2 r_1   $ &$   w_5 =  r_1 r_2 r_0 w_1 r_0 r_2 r_1    $ \\
\end{tabular}
\end{center}

\begin{table}[htb]
\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|c|c|} \hline
 & $z_1$  &$z_2$  & $z_3 $ & $z_4$ &$z_5$ & $w_1 $ &$w_2 $ &$w_3$  & $w_4$ & $w_5$ \\ \hline \hline
$r_0$ & $z_2$ & $z_1$ & $z_4$ & $z_3$ & $z_5^{-1}$ & $w_2$ & $w_1$ & $w_4$ & $w_3$ & $w_5$ \\ \hline
$r_1$ & $w_2^{-1} $ & $w_1$ & $z_3^{-1} $ & $z_5$ & $z_4$ & $z_2$ & $z_1^{-1} $ & $w_3$ & $w_5$ & $w_4$ \\ \hline
$r_2$ & $z_3$ & $z_4$ & $z_1$ & $z_2$ & $z_5$ & $w_3$ & $w_4$ & $w_1$ & $w_2$ & $w_5^{-1}$ \\ \hline
\end{tabular}
\caption{The relations of type (\ref{eq:normalize}) for (3.3.4.3.4)}
\end{center}
\end{table}


\subsection{(3.3.3.4.4)}
\begin{center}
\begin{tabular}{ll}
\noindent $z_1=[(r_0r_1)^3r_2(r_0r_1)^2r_2r_0r_1r_2(r_1r_0)^2r_2r_1r_2r_1r_2]^2$  \hspace{.5cm} &

$w_1= r_2 r_1 r_0 r_1 r_2 z_1 r_2 r_1 r_0 r_1 r_2$\\
$z_2 = r_1 z_1 r_1   $ &$   w_2 = r_1 w_1 r_1   $ \\
$z_3 = r_0 r_1 z_1 r_1 r_0   $ &$   w_3 = r_0 r_1 w_1 r_1 r_0   $ \\
$z_4 = r_2  z_1 r_2   $ &$   w_4 =  r_2 w_1 r_2    $ \\
$z_5 = r_1 r_2  z_1  r_2 r_1   $ &$   w_5 =  r_1 r_2 w_1 r_2 r_1    $ \\
\end{tabular}
\end{center}

\begin{table}[htb]
\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|c|c|} \hline
 & $z_1$  &$z_2$  & $z_3 $ & $z_4$ &$z_5$ & $w_1 $ &$w_2 $ &$w_3$  & $w_4$ & $w_5$ \\ \hline \hline
$r_0$ & $w_1^{-1}$ & $z_3$ & $z_2$ & $w_4^{-1} $ & $w_5$ & $z_1^{-1} $ & $w_3$ & $w_2$ & $z_4^{-1} $ & $z_5$\\ \hline
$r_1$ & $z_2$ & $z_1$ & $w_3^{-1} $ & $z_5$ & $z_4$ & $w_2$ & $w_1$ & $z_3^{-1} $ & $w_5$ & $w_4$\\ \hline
$r_2$ & $z_4$ & $z_3^{-1} $ & $z_2^{-1} $ & $z_1$ & $w_5^{-1}$ &  $w_4$ & $w_3^{-1} $ & $w_2^{-1} $ & $w_1$ & $z_5^{-1}$ \\ \hline
\end{tabular}
\caption{The relations of type (\ref{eq:normalize}) for (3.3.3.4.4)}
\end{center}
\end{table}

\subsection{(4.6.12)}
\begin{center}
\begin{tabular}{ll}
\noindent $z_1=(r_0r_1)^4[r_2(r_0r_1)^{-4}]^2r_2(r_0r_1)^6r_2[(r_0r_1)^4r_2]^2(r_0r_1)^{-4}r_2(r_0r_1)^6r_2$
&
$w_1= r_1 r_0 r_1 z_1 r_1 r_0 r_1$\\
$z_2 = r_1 z_1 r_1   $ &$   w_2 = r_1 w_1 r_1   $ \\
$z_3 = r_2 z_1 r_2   $ &$   w_3 = r_2 w_1 r_2   $ \\
$z_4 = r_1 r_2 z_1 r_2 r_1   $ &$   w_4 =  r_1 r_2 w_1 r_2 r_1    $ \\
$z_5 = r_2 r_1 z_1 r_1 r_2   $ &$   w_5 =  r_2 r_1 w_1 r_1 r_2    $ \\
$z_6 = r_2 r_1 r_2 z_1 r_2 r_1 r_2   $ &$   w_6 =  r_2 r_1 r_2 w_1 r_2 r_1 r_2    $ \\
\end{tabular}
\end{center}

\begin{table}[htb]
\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|c|c|c|c|} \hline
 & $z_1$  &$z_2$  & $z_3 $ & $z_4$ &$z_5$ & $z_6$ & $w_1 $ &$w_2 $ &$w_3$  & $w_4$ & $w_5$ & $w_6$ \\ \hline \hline
$r_0$ & $z_1$ & $w_2$ & $z_3$ & $w_4^{-1} $ & $w_5$ & $w_6^{-1}$ & $z_1^{-1} w_1^{-1} $ & $z_2$ & $z_3^{-1} w_3^{-1} $ & $z_4^{-1} $ & $z_5$ & $z_6^{-1}$ \\ \hline
$r_1$ & $z_2$ & $z_1$ & $z_4$ & $z_3$ & $z_6$ & $z_5$ & $w_2$ & $w_1$ & $w_4$ & $w_3$ & $w_6$ & $w_5$ \\ \hline
$r_2$ & $z_3$ & $z_5$ & $z_1$ & $z_6$ & $z_2$ & $z_4$ & $w_3$ & $w_5$ & $w_1$ & $w_6$ & $w_2$ & $w_4$ \\ \hline
\end{tabular}
\caption{The relations of type (\ref{eq:normalize}) for (4.6.12)}
\end{center}
\end{table}


\subsection{(3.3.3.3.6)}

\begin{center}
\begin{tabular}{ll}
\noindent $z_1=[(r_0r_1)^3r_2r_1r_2r_0r_1r_2]^4$ \hspace{.5cm} &

$w_1=r_1 r_0 z_1 r_0 r_1$\\
$z_2 = r_0 z_1 r_0   $ &$   w_2 = r_0 w_1 r_0   $ \\
$z_3 = r_2 z_1 r_2   $ &$   w_3 = r_2 w_1 r_2   $ \\
$z_4 = r_2 r_0 z_1 r_0 r_2   $ &$   w_4 =  r_2 r_0 w_1 r_0 r_2    $ \\
$z_5 = r_1 r_2  z_1  r_2 r_1   $ &$   w_5 =  r_1 r_2  w_1  r_2 r_1    $ \\
$z_6 = r_1 r_2 r_0 z_1 r_0 r_2 r_1   $ &$   w_6 =  r_1 r_2 r_0 w_1 r_0 r_2 r_1    $ \\
$z_7 = r_0 r_1 r_2 z_1 r_2 r_1 r_0   $ &$   w_7 =  r_0 r_1 r_2 w_1 r_2 r_1 r_0    $ \\
$z_8= r_0 r_1 r_2 r_0 z_1 r_0 r_2 r_1 r_0   $ &$   w_8 =  r_0 r_1 r_2 r_0 w_1 r_0 r_2 r_1 r_0    $ \\
$z_9 = r_2 r_1 r_2 r_0 z_1 r_0 r_2 r_1 r_2   $ &$   w_9 =  r_2 r_1 r_2 r_0 w_1 r_0 r_2 r_1 r_2    $ \\
$z_{10} = r_0 r_2 r_1 r_2 r_0 z_1 r_0 r_2 r_1 r_2 r_0  $ &$   w_{10} =  r_0 r_2 r_1 r_2 r_0 w_1 r_0 r_2 r_1 r_2 r_0    $ \\
\end{tabular}
\end{center}

\begin{table}[htb]
\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|c|c|} \hline
 & $z_1$  &$z_2$  & $z_3 $ & $z_4$ &$z_5$ & $z_6 $ &$z_7 $ &$z_8$  & $z_9$ & $z_{10}$ \\ \hline \hline
$r_0$ & $z_2$ & $z_1$ & $z_4$ & $z_3$ & $z_7$ & $z_8$ & $z_5$ & $z_6$ & $z_{10}$ & $z_9$ \\ \hline
$r_1$ &$z_2w_2^{-1} $ & $w_1$ & $z_5$ & $z_6$ & $z_3$ & $z_4$ & $z_8$ & $z_7$ & $w_{10}^{-1} $ & $z_9^{-1} w_9$ \\ \hline
$r_2$ &$z_3$ & $z_4$ & $z_1$ & $z_2$ & $z_7^{-1} $ & $z_9$ & $z_5^{-1} $ & $z_{10}$ & $z_6$ & $z_8$ \\ \hline
\end{tabular}
\caption{Relations of type (\ref{eq:normalize}) for (3.3.3.3.6)}
\end{center}
\end{table}

\begin{table}[htb]
\begin{center}
\begin{tabular}{|c||c|c|c|c|c|c|c|c|c|c|} \hline
 & $w_1$  &$w_2$  & $w_3 $ & $w_4$ &$w_5$ & $w_6 $ &$w_7 $ &$w_8$  & $w_9$ & $w_{10}$ \\ \hline \hline
$r_0$ & $w_2$ & $w_1$ & $w_4$ & $w_3$ & $w_7$ & $w_8$ & $w_5$ & $w_6$ & $w_{10}$ & $w_9$ \\ \hline
$r_1$ & $z_2$ & $z_1^{-1} w_1 $ & $w_5$ & $w_6$ & $w_3$ & $w_4$ & $w_8$ & $w_7$ & $z_{10}w_{10}^{-1}$ & $z_9^{-1}$ \\ \hline
$r_2$ &$w_3$ & $w_4$ & $w_1$ & $w_2$ & $w_7^{-1} $ & $w_9$ & $w_5^{-1} $ & $w_{10}$ & $w_6$ & $w_8$ \\ \hline
\end{tabular}
\caption{Remaining relations of type (\ref{eq:normalize}) for (3.3.3.3.6)}
\end{center}
\end{table}

\end{appendices}
\end{document}
